# CORPORATE GOVERNANCE MECHANISMS AND FIRM FINANCIAL PERFORMANCE IN NIGERIA

Ahmadu Sanda, Aminu S. Mikailu and Tukur Garba

AFRICAN ECONOMIC RESEARCH CONSORTIUM

CONSORTIUM POUR LA RECHERCHE ECONOMIQUE EN AFRIQUE

# Corporate governance mechanisms and firm financial performance in Nigeria

By I

Ahmadu Sanda,
Aminu S. Mikailu
and
Tukur Garba
Department of Economics
Usmanu Danfodiyo University,
Sokoto, Nigeria

AERC Research Paper 149
African Economic Research Consortium, Nairobi
March 2005

© 2005, African Economic Research Consortium.

Published by: The African Economic Research Consortium

P.O. Box 62882-00200 Nairobi, Kenya

Printed by: The Regal Press Kenya, Ltd.

P.O.Box 46166-00100

Nairobi, Kenya

# **Table of Contents**

List of tables Acknowledgements Abstract

| 1.  | Introduction   | 1  |
|-----|--|----|
| 2.  | Purpose and objectives of the study                      | 3  |
| 3.  | The operating environment of the Nigerian stock exchange | 5  |
| 4.  | Theoretical framework                                    | 7  |
| 5.  | Literature review  | 9  |
| 6.  | Methodology  | 13 |
| 7.  | Results  | 18 |
| 8.  | Concluding remarks                                       | 30 |
| 9.  | Note   | 32 |
| 10. | Refernce   | 33 |
| 11. | Appendix: List of companies used in the sample           | 36 |

# List of tables

| 1. | Variable definitions and measurement | 14 |
|----|--------------------------------------|----|
| 2. | Coefficient estimates for Equation 1 | 19 |
| 3. | Coefficient estimates for Equation 2 | 20 |
| 4. | Coefficient estimates for Equation 3 | 21 |
| 5. | Coefficient estimates for Equation 4 | 22 |
| 6. | Coefficient estimates for Equation 5 | 23 |
| 7. | Coefficient estimates for Equation 6 | 25 |
| 8. | Coefficient estimates for Equation 7 | 26 |
| 9. | Coefficient estimates for Equation 8 | 28 |

# **Acknowledgements**

A number of institutions and individuals have contributed to the undertaking of this study. We are grateful to the African Economic Research Consortium for sponsoring the study under its thematic research grant scheme, and to its staff for the very efficient way they facilitated the research. We owe a great deal of gratitude to the Usmanu Danfodiyo University Sokoto for providing office, secretarial and logistical support. We acknowledge the support of the Central Bank of Nigeria, which in 2001 offered financial assistance that provided opportunity for the development of a database and for the collection of material that culminated in generating ideas for developing a proposal for this research.

We thank the participants at the December 2001, May 2002 and November/December 2003 AERC biannual research workshops held in Nairobi. The deliberations at the workshops and in particular those held by the Finance and Resource Mobilization Group were very illuminating, helping to provide useful suggestions and comments for improving the research. In this regard, particular gratitude is due to Christopher Adam, Ernest Aryeetey, Paul Collier, Lawrence Harris, Allechi M'bet, Leonce Ndikumana, Machiko Nissanke, Catherine Pattillo and Lemma Senbet. We are, however, solely responsible for any errors in the study.

Last but by no means the least, we are grateful to Rakkiya, Bilkisu, Salamatu, Hussainatu and Ibrahim and all our loved ones for their support, encouragement and understanding.

#### **Abstract**

Recent global events concerning high-profile corporate failures have put back on the policy agenda and intensified debate on the efficacy of corporate governance mechanisms as a means of increasing firm financial performance. This study attempts to address this question using pooled ordinary least squares regression analysis for a sample of 93 firms quoted on the Nigerian Stock Exchange for the period 1996-1999. While making a case for a board size of ten and for concentrated as opposed to diffused equity ownership, the results argue for the separation of the posts of Chief Executive Officer (CEO) and Chair. Moreover, although the results find no evidence to support the idea that boards with a higher proportion of outside directors perform better than other firms, there is evidence that firms run by expatriate CEOs tend to achieve higher levels of performance than those run by indigenous CEOs. In the main, the results are consistent with existing literature, but there is need to err on the side of caution in any attempt to generalize the findings as the sample selection was determined by the availability of data rather than by any probability criterion.

Key words: corporate governance; agency theory; stakeholder theory

#### 1. Introduction

orporate governance is concerned with ways in which all parties interested in the well-being of the firm (the stakeholders) attempt to ensure that managers and other insiders take measures or adopt mechanisms that safeguard the interests of the stakeholders. Such measures are necessitated by the separation of ownership from management, an increasingly vital feature of the modern firm. A typical firm is characterized by numerous owners having no management function, and managers with no equity interest in the firm. Shareholders, or owners of equity, are generally large in number, and an average shareholder controls a minute proportion of the shares of the firm. This gives rise to the tendency for such a shareholder to take no interest in the monitoring of managers, who, left to themselves, may pursue interests different from those of the owners of equity. For example, the managers might take steps to increase the size of the firm and, often, their pay, although that may not necessarily raise the firm's profit, the major concern of the shareholder.

Financial economists have long been concerned with ways to address this problem, which arises from the incongruence of the interests of the equity owners and managers, and have conducted significant research towards resolving it. The literature emanating from such efforts has grown, and much of the econometric evidence has been built on the theoretical works of Ross (1973), Jensen and Meckling (1976), and Fama (1980). At the initial levels of the development of the theory of agency, especially as it relates to the firm, concern seemed to focus more on the relationship between the management and shareholders than between them and other categories of stakeholders. The stakeholder theory has of late captured the attention of researchers and a survey of literature on this aspect of corporate finance can be found in the works of John and Senbet (1998). According to this theory, the firm can be considered as a nexus of contracts between management on the one hand and employees, shareholders, creditors, government and all other stakeholders on the other. Thus, from the point of view of the stakeholder theory, concern should go beyond the traditional management-shareholder relationship to include all other stakeholders such as mentioned above. The stakeholder theory has undergone some refinements in the work of Jensen (2001), who presents what he terms the "enlightened stakeholder theory". For him, the traditional stakeholder theory encourages managers to be servants of many masters, with no clear guidance whenever trade-offs (or indeed, conflicts) occur, as they often do. He argues that the absence of any criterion for choice in cases of trade-offs (or conflicts) tends to give managers some discretionary powers to serve the master of their own choice. As we will see in a subsequent section, Jensen proposes a single criterion - addition to the long-term value of the firm - for

1

managers to pursue so that the interests of all key stakeholders can be served. This is based on the idea that changes in the long-term value of the firm would be difficult to materialize if the interest of a key stakeholder were not protected.

Empirical work in the area of corporate governance has undergone a remarkable growth, founded mostly on the basis of management-shareholder conflict and to a lesser but increasing extent on the stakeholder theory. Despite the volume of empirical evidence, there has been no consensus on how to resolve the problem. The lack of consensus has produced a variety of ideas (or mechanisms) on how to deal with the problem of agency. The mechanisms we are concerned with in this study can be divided into five: striking a balance between outside and inside directors; promoting insider (i.e., managers and directors) shareholding; keeping the size of the board reasonably low; encouraging ownership concentration; and encouraging the firm to have a reasonable amount of leverage in the expectation that creditors might take on a monitoring role in the firm in order to protect their debt holdings. These mechanisms are detailed in Section 6 of this paper.

To achieve the stated objective, this paper is structured into eight sections. After the introduction, we provide in Section 2 the justification of the study and its objectives. Section 3 gives an overview of the regulatory environment in which the Nigerian Stock Market operates, emphasizing the link between corporate governance and a weak stock market saddled with severe problems of low levels of liquidity and trading volumes. Section 4 is the theoretical framework for the study, while the literature review in Section 5 provides a survey of stylized facts emerging from earlier theoretical and empirical works. Section 6 details the methodology and Section 7 presents the results. The last section offers some conclusions.

# 2. Purpose and objectives of the study

his research is an attempt to examine the extent to which the suggested mechanisms might help reduce the agency problem in a developing stock exchange such as that of Nigeria, where there is a yawning gap between theory and evidence. We are aware of no published work in this area on the Nigerian Stock Exchange (NSE). The only unpublished work from Nigeria is a recently completed AERC-supported research by Adenikinju and Ayorinde (2001), who used data on the NSE to examine the relationship between firm performance and two of the five mechanisms listed above - insider ownership and ownership concentration. The authors reported no significant relationship between firm performance, on the one hand, and ownership concentration and managerial shareholding, on the other. Although helping to shed more light on ways to reduce the agency problem in Nigeria, Adenikinju and Ayorinde (2001) suffers from the weakness of excluding important mechanisms for addressing the agency problem. There is also a methodological loophole as their measure of ownership concentration does not allow for inter-firm comparison as the number of concentrated shareholders varied from firm to firm in their sample.

# Purpose of the study

By attempting to address an important limitation of Adenikinju and Ayorinde (2001), this study aims to provide additional insights into the relationship between governance mechanisms and firm financial performance in Nigeria. Our focus is on the five dimensions of corporate governance, abstracting from other dimensions such as incentive schemes. It is hoped that the evidence would serve as important quantitative information into the cauldron of policy as well as add to the existing body of empirical literature from a developing stock exchange such as that of Nigeria. The need for a study of this kind is even more important in an environment like Nigeria's, which is characterized by growing calls for effective corporate governance, particularly for public limited liability companies. This call is understandable in view of the importance of effective governance at both microeconomic and economy-wide levels.

At the level of the firm, it offers the promise of a fair return on capital invested through improved efficiency (Metrick and Ishii, 2002). It also has some implications for the ongoing privatization programme that the Government of Nigeria is currently undertaking. Grosfeld (2002), citing the works of other scholars, indicated that the effectiveness of privatization is greater when corporate governance works well. Moreover,

by helping to promote firm performance and the protection of stakeholder interest, corporate governance encourages investment and stock market development, which Demirguc-Kunt and Levine, (1996) have associated with improved macroeconomic growth. Further, recent evidence in the work of Klapper and Love (2002) suggests that firm-level corporate governance provisions matter more in countries with weak legal (or regulatory) environments, implying that "firms can partially compensate for ineffective laws and enforcement by establishing good corporate governance and providing credible investor protection" (Abstract).

# Study objectives

Given the overall objective of examining the relationship between corporate governance mechanisms and firm financial performance in Nigeria, this study had several specific objectives. In particular the study sought to:

- Examine the extent to which insider shareholding may be related to firm financial performance;
- ° Ascertain the influence of the composition of board members on firm performance;
- Investigate the relationship between board size and firm performance;
- Assess the influence of block holdings or ownership concentration on firm performance;
- Examine whether or not the separation of the posts of CEO and Board Chair is of any value in the promotion of firm performance;
- Examine whether or not the appointment of an expatriate CEO has any role to play in the promotion of stakeholder interest; and
- Examine whether, within a certain range, a positive relationship exists between debt and firm performance.

In line with these objectives, the hypotheses to be tested in this study are as well divided into seven, as follows:

- There is a positive relationship between insider shareholding and firm performance.
- There is a positive relationship between the proportion of outside directors on the board and the performance of the firm.
- There is a significant relationship between firm performance and the size of the board
- There is a positive relationship between ownership concentration and firm performance.
- Firms in which the posts of CEO and Chair are separated tend to perform better than those with a combined role for the two posts.
- Firms with expatriate CEOs tend to perform better than those with indigenous CEOs.
- Within a certain range of leverage, a positive relationship exists between debt and firm performance.

# 3. The operating environment of the Nigerian Stock Exchange

he Nigerian Stock Exchange, which until 1977 was known as the Lagos Stock Exchange, came into being in 1960, but started operations with less than ten stocks in 1961. At age 43 in 2003, the exchange boasted about 200 stocks, quite a remarkable growth rate considering the number at the initial stage, but well below the figure of over 600 in each of the Malaysian, South African and South Korean exchanges. The discrepancy is even more pronounced with respect to the market capitalization of these other stock exchanges. According to Standard and Poor's is (2000), as at 1999, the market capitalization for the Nigerian Stock Exchange stood at US\$2.94 billion. This compares with US\$145,445 billion for the Malaysian exchange, US\$262,478 billion for South Africa's and US\$308,534 billion for South Korea's. In relative terms, as at 1999, the market capitalization of the Nigerian stock exchange was equivalent to only 2%, 1.1% and 0.9% of that of the Malaysian, South African and South Korean stock exchanges, respectively. In fact, if account were taken of the fact that the 20 largest stocks on the NSE together account for 73.8% of the market capitalization (Nigerian Stock Exchange, 2001), the very small size of most of the stocks on the NSE would be better appreciated.

The small number and value of stocks listed on the exchange are probably a manifestation of severe problems affecting the exchange. Such problems include a serious liquidity problem, low demand for securities and low trading volume. Liquidity of a stock exchange is concerned with the ease of trading in shares on the market. A liquid stock exchange allows for easier access to debt (debenture) and bank loans (through the use of stocks as collaterals). Writing on the Nigerian Stock Exchange, Emenuga (1998) noted that the liquidity of the market averaged just 2%, well below the average for many African bourses, and a very far cry from the average for Taiwan (174.9%) and South Korea (97.8%)! These problems may not be unrelated to the weak regulatory environment in which the market has operated.

Although the stock exchange began operations in 1960, a regulatory body, the Securities and Exchange Commission (SEC), was established almost two decades later, in 1979. It took another two decades for the Securities and Investment Act (1999) to come into being. The Act was the first comprehensive legal document providing rules and regulations for the conduct of operators in the exchange. Thus, the stock exchange operated for almost two decades without a regulatory organ, and for another two with a regulatory organ weakened by the absence of a comprehensive legal framework for the discharge of its regulatory duties. Emenuga (1998) gives an account of some of the shady practices that prevailed in the exchange, and it may not be an overstatement to link at least some of them to a weak regulatory environment.

Previous research has suggested a link between corporate governance and the development of a legal/regulatory environment. Klapper and Love (2002), for example, report that firms in countries with weak overall legal systems have on average lower governance rankings. In addition, they find that good governance is positively correlated with market valuation and operating performance, which implies a positive correlation between the effectiveness of the regulatory environment and the performance of firms. They also observe that in countries with weak laws the degree of flexibility of firms to affect their own governance is likely to be smaller (i.e., the firm is likely to be constrained by the country-level legal provisions). Garcia and Liu (1999) have also attempted to provide a link between the regulatory environment and governance mechanisms. A regulatory environment that encourages mandatory disclosure of reliable information about firms may enhance investor participation. Moreover, regulations that instil investors' confidence in brokers have the capacity to encourage investment and trading in the stock exchange. In a study on the Ukrainian stock exchange, Dean and Andreyeva (2001) report that the regulatory environment can in fact have an important consequence for the kind of governance structures that emerge. The authors found that the weak regulatory and legal environment of Ukr aine tended to favour concentrated over diffused ownership structures.

# 4. Theoretical framework

he theoretical framework upon which this study is based is the agency theory, which posits that in the presence of information asymmetry the agent (in this case, the directors and managers) is likely to pursue interests that may hurt the principal, or shareholder (Ross, 1973; Fama, 1980). At first the theory was applied to the relationship between managers and equity holders with no explicit recognition of other parties interested in the well-being of the firm. Subsequent research efforts widened the scope to include not just the equity holders but all other stakeholders, including employees, creditors, government, etc. This approach, which attempts to align the interests of managers and all stakeholders, has come to be regarded as the stakeholder theory.

The stakeholder theory has been a subject of some investigation. John and Senbet (1998) provide a comprehensive review of corporate governance, with a particular focus on the stakeholder theory. The authors note the presence of many parties interested in the well-being of the firm and that these parties often have competing interests. While equity holders might welcome investments in high yielding but risky projects, for example, such investments might jeopardize the interests of debt holders especially when the firm is teetering on the edge of bankruptcy. The review also emphasizes the role of non-market mechanisms, citing as an example the need to determine an optimal size of the board of directors especially in view of the tendency for board size to exhibit a negative correlation with firm performance. Other non-market mechanisms reviewed by John and Senbet include the need to design a committee structure in a way that allows the setting up of specialized committees with different membership on separate critical areas of operations of the firm. Such a structure would allow, for example, productivity-oriented committees and monitoring-oriented ones.

In an article extending the stakeholder theory, Jensen (2001) also recognizes the multiplicity of stakeholders. He concurs with John and Senbet that certain actions of management might have conflicting effects on various classes of stakeholders. This implies that the managers have a multiplicity of objective functions to optimize, something that Jensen sees as an important weakness of the stakeholder theory "because it violates the proposition that a single-valued objective is a prerequisite for purposeful or rational behaviour by any organisation" (Jensen, 2001: 10). In search of a single valued objective function that conforms with rationality, Jensen suggests a refinement of the stakeholder theory - the enlightened stakeholder theory. For him, the enlightened stakeholder theory offers at least two advantages. First, unlike the earlier version with multiple objectives, the modified form of the theory proposes only one objective that managers should pursue: the maximization of the long-run value of the firm. If the interest of any major stakeholder

was not protected, the objective of long-run value maximization would not be achieved. A second, related, appeal of the enlightened stakeholder theory is that it offers a simple criterion to enable managers to decide whether they are protecting the interests of all stakeholders: invest a dollar of the firm's resources as long as that will increase by at least one dollar the long-term value of the firm. There is an important caveat, however. Jensen himself cautions that the criterion may be weakened by the presence of a monopoly situation or externalities.

Despite its appeal, the stakeholder theory of the variety proposed by Jensen has not been subjected to much empirical evaluation. At least two factors might have contributed to the gap between theory and evidence. The first, already alluded to, concerns the prevalence of externalities and monopoly situation. The second is the problem of measurement, especially in view of the problems associated with getting an accurate measure of the long-term value of the firm.

#### 5- Literature review

he literature suggests that both market and non-market mechanisms could be used to promote the alignment of interest of managers and stakeholders. The managerial labour market and the market for corporate takeover tend to exert pressures both within and outside the firm in order to achieve such an alignment of interest. Fama (1980) asserts that a firm can be viewed as a team, whose members realize that in order for the team to survive, they must compete with other teams, and that the productivity of each member has a direct effect on the team and its members. Thus, within the firm, each manager has the *incentive to monitor* the behaviour of other managers, whether subordinates or superiors. Secondly, Fama (1980) argues that the firm is in the market for new managers and the reward system must be based on performance in order for it to attract good managers or even to retain existing ones.

Demsetz and Lehn (1985) provide an explanation for the weakness of the market-induced mechanisms as a means of protecting stakeholder interests. They observe that the free rider problem tends to prevent any of the numerous owners of equity from bearing the cost of monitoring the managers.

Empirical works abound on the mechanisms aimed to help reduce the agency problem. Abstracting from other dimensions of corporate governance (such as incentive schemes) we focus on five mechanisms - insider shareholding, board composition, board size, ownership concentration and debt.

# Insider shareholding and firm value

The first argument to address the problem of agency concerns the use of insider shareholding. Several researchers (DeAngelo andDeAngelo, 1985; McConnell and Servaes, 1990; Loderer and Martin, 1997; Nor et al., 1999; Yeboah-Duah, 1993) have undertaken research on this aspect, reporting very conflicting results. In particular, McConnell and Servaes (1990) find a significant curvilinear relationship between insider ownership and firm performance. While Loderer and Martin (1997) find no significant relationship, Nor et al. (1999) reported a non-linear relationship, drawing conclusions contrary to those of Yeboah-Duah (1993).

# Composition of board members

The composition of board members is also proposed to help reduce the agency problem (Weisbach, 1988; Hermalin and Weisbach, 1991). A positive relationship is expected between firm performance and the proportion of outside directors sitting on the board. Unlike inside directors, outside directors are better able to challenge the CEOs. It is perhaps in recognition of the role of outside directors that in the UK a minimum of three outside directors is required on the board; in the US, the regulation requires that they constitute at least two-thirds of the board (Bhagat and Black, 2001).

Empirical evidence has grown but the results are very conflicting. Studies by Weisbach (1988), Mehran (1995) and Pinteris (2002) have produced evidence in support of a positive role for outside directors on firm performance. John and Senbet (1998) in a survey of corporate governance reported that the work of Fosberg (1989) was in support of this positive role.

Other works have reported no evidence of a significant relationship between firm performance and the proportion of outside directors on the board (Bhagat and Black, 1999,2000; Hermalin and Weisbach, 1991; Yermack, 1996; and Metrick and Ishii, 2002). In fact Weir and Laing (2001) reported a negative relationship!

John and Senbet (1998) stress the role of committee structure as a means of increasing the independence of the board. They refer to the work of Klein (1998) and argue for the need to set up specialized committees on audit, remuneration and appointment.

Unlike the preceding argument in support of board structures, Laing and Weir (1999) play down their importance, stressing instead the importance of business experience and entrepreneurship. According to them, firms managed by dynamic CEOs tend to perform better than other categories of firms. On the assumption that foreign firms are managed by more experienced CEOs, Estrin et al. (2001) test whether foreign firms perform better than domestic ones in Bulgaria, Romania and Poland. Using panel data for the three countries for the period 1994-1998, they find x\*at irrespective of the estimation technique, foreign firms perform better than private domestic firms. They attribute this finding to the possibility that foreign firms might have some superior knowledge, which leads them to be more efficient. A common theme running through the two studies is the important role that the experience and skills of chief executives could play as a means for improving firm performance.

#### Board size

The third mechanism proposed to deal with the agency problem is board size. There are arguments in favour of small board size. First, Yermack (1996), in a review of the earlier work of Monks and Minow (1995), argues that large boardrooms tend to be slow in making decisions, and hence can be an obstacle to change. A second reason for the support for small board size is that directors rarely criticize the policies of top managers and that this problem tends to increase with the number of directors (Yermack, 1996; Lipton and Lorsch, 1992).

Yermack (1996) examines the relation between board size and firm performance, **concluding** that the smaller the board size the better the performance, and proposing an optimal board size of ten or fewer. John and Senbet (1998) maintain that the findings of Yermack have important implications, not least because they may call for the need to depend on forces outside the market system in order to determine the size of the board.

#### Block holdings or ownership concentration

The fourth element of governance mechanism examined in this study is ownership concentration, which refers to the proportion of a firm's shares owned by a given number of the largest shareholders. A high concentration of shares tends to create more pressure on managers to behave in ways that are value-maximizing. In support of this argument, Gorton and Schmid (1996), Shleifer and Vishny (1997), Morck et al. (1988), and Wruck (1989) suggest that at low levels of ownership concentration, an increase in concentration will be associated with an increase in firm value, but that beyond a certain level of concentration, the relationship might be negative.

Other studies such Renneboog (2000) reported results not totally in agreement with the hypothesis of a positive relationship. Using a set of variables suggested by Agrawal and Knoeber (1996), the author reported no evidence to support the hypothesis of a positive relationship between firm performance and ownership concentration. Holderness and Sheehan (1988) find little evidence that high ownership concentration directly affects performance.

#### The Role of Debt

Pinally, debt owed to large creditors such as banks is also believed to be a useful tool for reducing the agency problem. Large creditors, like large stakeholders, also have interest in seeing that managers take performance-improving measures. Empirical evidence seems to be in support of this assertion. Shleifer and Vishny (1997) in a review article, cite the works of Kaplan and Minton (1994) and Kang and Shivdasani (1995), who found higher incidence of management turnover in Japan in response to poor performance in companies that have a principal banking relationship relative to companies that do not.

Another form of agency problem, known as debt agency, arises when there is a conflict of interests between stockholders and debt holders. Debt holders are entitled to claims and these have the tendency to rise at low levels of firm performance, and to remain constant beyond a certain level of that performance. Thus, good performance benefits the stockholders more than it does debt holders, but this is not true when performance is very low. In fact, as the firm moves towards bankruptcy, equity holders face the risk of losing only their shareholdings, passing the burden of such bankruptcy to the debt holders. Taken together, these outcomes encourage managers working to protect the interest of equity holders to embark on risky, high-return projects. This could lead to economic

inefficiency since "projects that are otherwise profitable may be foregone [sic] in exchange for high risk but inferior counterparts" (John and Senbet, 1998: 378).

The literature seems to present no unanimous position on the role of debt. **Although** some see it as having the potential to induce the right steps by the board to **protect** shareholder interests, other scholars point to the emergence of debt agency and to the need to constitute boards in ways that would protect both shareholder and creditor interests. To achieve this, it is suggested that the board should have a representation from the creditors, as is often the case in Japan and Germany where banks have significant debt holding interests.

# 6. Methodology

he data used for this study were derived from a number of sources. Data on directors' shareholding were obtained from the database of a Lagos-based stock broking firm, which provided for each firm used in the sample, a list of directors, the number of shares of the firm owned by each director and the number of outstanding shares. The second set of data was extracted from Nigerian Stock Exchange (2000). This set comprises the value of total assets, value of share capital and earnings for the year, as measured by profit after tax. Information on board size and board composition was obtained from Genmax (1998). This source also provided information on the status of CEOs, whether Nigerian or expatriate. The annual report of the Securities and Exchange Commission of Nigeria provided for each firm year-end market capitalization and the price-earnings (PE) ratio. The sample covered the period 1996 through 1999. This period coincides with the computerization of records and the associated increase in the reliability and availability of data. In all, 180 companies were listed on the NSE at the time of data collection. A non-probability sampling technique was adopted as only firms with the required information were selected in the study. Table 1 provides a list of variables, their definitions and method of computation. The Appendix lists the sample used in this study of 93 firms drawn from the 14 sectors of the exchange.

Unlike Adenikinju and Ayorinde (2001), who did not include banking and insurance firms in their sample and analysis, we did include financial firms and analysed them together with other firms. There were two reasons for doing so. First, we assumed that including them would increase our sample size, which might lead to better results. Second, we included dummy variables in our model for the sectors included in our sample to control for sectoral variations.

Data on ownership concentration were not available, so a proxy for it had to be calculated. Genmax reported data on the proportion of shares owned by the largest shareholders for each of the firms in the sample. To be useful for our purposes, the data had to be refined as the number of largest shareholders varied across firms, making inter-firm comparison difficult. To overcome this weakness the proportion of shares owned by the largest shareholders was divided by the number of largest shareholders. This provided a crude (but so far the best) measure of ownership concentration.

In all, a total of eight equations were estimated. Equation 1 specifies four independent variables: director shareholding (DIRSHARE), number of directors on the board (BOARDSIZE), the proportion of outside directors sitting on the board (OUTSIDE) and the extent of ownership concentration (CONCENT).

Table 1: Variable definitions and measurement

| Variable        | Definition                               | Measurement   |
|-----------------|--|---|
| PE Ratio        | Price-Earning ratio                      | Ratio of share price to earning per street.   |
| ROA             | Returns on assets                        | Net profit as a prerpantage of the rotal assets.  |
| ROE             | Returns on equity                        | Net proid as a percentage of equity you're.   |
| Tobin's Q       | Modified Tobin's Q                       | Year-endmake appointain build by the book value of the market value of end year and include a value of debt divided by the book refue of total assets.              |
| DIRSHARE        | Directors shareholding                   | Total number of weares person to suspens of a given from one personnage of the outstanding shares of the implient percentage the greater the director shareholding. |
| BOARDSIZE       | Board size                               | Number of discolors on the based.   |
| OUTSIDE         | Number of outside directors on the board | Proportion of caterial directions withing an the board.   |
| CONCENT         | Ownership concentration                  | The proportion of thems as weed by the included shareholders divided by the countries of largest shareholders.  |
| Debt            | Leverage                                 | The ratio of distributions applicate  |
| FIRMSIZE        | Firm size in terms of total assets owned | The natural ចំណាមរយៈ នេះគាំ   |
| CEOSTATUS       | Role of CEO                              | A dummy variable, taking a value of \$100 lims with CEO as Chair, and inclinaviers.   |
| DIRSHSQUARE     | Quadratic term                           | Square of directors shere with the same.  |
| BDSIZESQUARE    | Quadratic term                           | Square of boardssize.   |
| CEOFOREIGN      | A firm that has a foreign CEO            | A dummy variable taking a value of Goodins with Nigerian CEEs, and it settle was a  |
| CONCENTSQ       | Quadratic term                           | Square of ownership concentration.  |
| SD <sub>p</sub> | Sectoral dummies                         | Dummy variables are all but the enterpoint of e sector in the same to which is taken as a base.   |

FIRMPERFORM. = 
$$a_0 + a_L DIRSHARE$$
. +  $a_2 BOARDSIZE + a_3 OUTSIDE_I + a_j CONCENT$ . +  $^{\wedge}$  (1)

Four alternative measures of firm performance (the dependent variable in Equation 1) were computed: ROA, ROE, PE ratio and Tobin's Q (hereafter, Q). Data on PE ratio were readily available so no additional computation was needed; the original Tobin's Q, named after James Tobin, is defined as the ratio of market value of debt and equity of the firm to the replacement cost of the firm (Nor et al., 1999). Replacement cost information could not be found in the Nigerian context as in Malaysia, however, so a modified form of Q was calculated by dividing year-end market capitalization by the book value of total assets. The modification was adopted from the works of Nor et al. (1999) on Malaysia and Demirguc-Kunt (1992). Estimates of all the parameters (except a<sub>2</sub>) in Equation 1 are expected to be positive. Equation 1 was therefore estimated three times: once each for the ROA, ROE and PE ratio. In the case of Q, ownership concentration was dropped from the equation as its inclusion led to unstable estimates.

Equation 2 was obtained by taking the natural logs of all but one variable (BOARDSIZE) in Equation 1 and estimated four times.

FIRMPERFORM. = 
$$(3_Q + b,DIRSHARE. + P_2BOARDSIZE + P_3OUTSIDE_{|} + ^CONCENT. + ji.$$
 (2)

Equation 3 was obtained by adding the natural logs of total assets to Equation 2 in order to control for firm size. Equation 3 was also estimated four times, once each for the four measures of firm performance.

FIRMPERFORM. = 
$$\diamondsuit_0$$
 + DIRS HARE. + ( $\triangleright_2$ BOARDSIZE. + ( $j^\circ$ OUTSIDE $j$  +  $^\circ$ CONCENT. ( $)_5$ FIRMsizE.+  $|x|$  (3)

The relationship between firm performance and certain governance variables has been reported in the literature to exhibit a non-linear relationship, tending to rise at low levels, achieve a maximum and then decline thereafter. There is also a debate concerning the need to separate the functions of CEO and Chair. To capture the effects of non-linearity as well as examine the effects of the separation of the roles of CEO and Chair, Equation 4 was obtained by adding to Equation 3 the squared values of *DIRSHARE* and *BOARDSIZE* as well as the status of CEO (CEOSTATUS), a dummy variable, taking a value of zero for firms with CEO Chair, and 1 otherwise. Although running a non-parametric estimate can capture non-linearity, it is not used in this study owing to the weakness of the estimate. Parametric estimates are considered more powerful compared with non-parametric ones (Pallant, 2004). Furthermore, non-parametric estimates may not detect differences or relationships, even when they actually exist (Pallant, 2004; Gupta, 1999).

FIRMPERFORM. = 
$$X_Q$$
 + ^OUTSIDE. + A,2DIRSHAREi +  $X_3$ DIRSHSQUARJE.+A,4B o ARDSIZE! + A,5CEOSTATUS. +  $X_3$ DIRSHSQUARE. + ^CONCENT. + \FIRMSIZE. + ^ (4)

Equation 4 was also estimated four times, one each for the four measures of firm performance.

As indicated in the literature review, some researchers such as Laing and Weir (1999) and Estrin et al. (2001) emphasize the importance of business experience and entrepreneurship especially for the chief executives. To examine the extent to which business experience and skills of CEOs play an important role in affecting firm performance, we made a simplifying assumption that in Nigeria, compared with their local counterparts, foreign CEOs have more experience in modern management techniques and have greater international exposure. Therefore, in order to test whether the business experience and skills of the CEO are related to firm performance, we divided the firms in the sample into those with foreign and those with local CEOs. Equation 5 was therefore obtained by adding a dummy variable, *CEOFOREIGN*, to Equation 4.

FIRMPERFORM. = 
$$5_0$$
 + d, OUTSIDE. +  $5_2$ DIRSHARE. +  $5_3$ DIRSHSQUARE, +  $5_4$ BOARDSIZE! +  $5_5$ CEOFOREIGN4- $5_6$ CEOSTATUS. +  $5_7$ BDSIZESQUARE. +  $5_8$ CONCENT. +  $8_9$ FERMSIZE. +  $^{\wedge}$  (5)

The dummy variable took a value of zero for Nigerian CEOs and 1 otherwise. Equation 5 was then estimated four times, once each for the four measures of firm performance.

The relationship between firm performance and ownership concentration has been suggested to take a non-linear form, tending to be positive only within a certain range of ownership concentration. To examine the relevance of this argument, Equation 6 was obtained by including a quadratic term in Equation 5.

FIRMPERFORM. = 
$$q_0$$
 + ^OUTSIDE. + ( $p_2$ DIRSHAREi + ( $p_3$ DIRSHSQUARE. + ( $p_4$ BOARDSIZE +  $< p_5$ CEOFOREIGN + cp<sub>6</sub>CEOSTATUS. + ( $p_7$ BDSIZESQUARE. + ( $p_g$ CONCENT. + ( $p_9$ COCENTSQ. + ( $p_{10}$ FIRMSIZE. + ja. ' ' (6)

Because the firms in the sample for this study were drawn from 14 different sectors of the exchange, there was need to incorporate sector dummies in order to account for risk differences. Equation 7 was therefore obtained by adding to Equation 6 a set of 13 sector dummy variables.

$$\begin{aligned} & FIRMPERFORM = B_0 + 0, OUTSIDE. + 6_2DIRSHAREi + \\ & B_3DIRSHSQUARE. + e_4BOARDSIZE. + 0_5CEOFOREIGN + \\ & OgCEOSTATUS. + 9_7BDSIZESQUARE. + 0_8CONCENT. \\ & + 0_9COCENTSQ_i + 0_{10}FTRMSIZE. + 1 a D . + (x \end{aligned} \tag{7}$$

where ZD..are 13 dummy variables for all but the automobile sector in the sample, i = 1,...,93, j = 11't...,23.

There are two major loopholes in the approach thus far. The first concerns the need to incorporate leverage into the computation of Tobin Q. In the absence of market value of debt, we utilize its book value to further compute a modified form of the Tobin Q. Therefore, Tobin's Q is further modified as the sum of the market value of equity and the book value of debt divided by the book value of total assets. This was adopted from the work of Oxelheim and Randoy (2001). The second loophole concerns the need to incorporate debt as a governance mechanism, requiring its inclusion amongst the set of regressors. Thus, Equation 8 was obtained by modifying Equation 7 to account for these two considerations.

#### 7. Results

he results for all these equations are divided into two types, descriptive **results** and those obtained from the regression analysis. The Statistical Package for the Social Sciences, SPSS, was used for both types of analysis.

# Descriptive statistics

The analysis begins by examining the basic features of the data using the descriptive statistics as a starting point. Averages were obtained for market capitalization, total assets and net profit. The firms in the sample reported an average market capitalization of N2.49 billion, equivalent to US\$29.29 million. The average value of total assets was computed at N6.78 billion (or US\$79.76 million), while that for net profit was found to be N637 million (or US\$7.49 million). Given that the sample contained a substantial proportion of large firms, the averages reported here are expected to be above the market average. Nonetheless, these measures of firm size are a clear pointer to the small size of the firms operating in the Nigerian Stock Exchange.

The analysis also examined data for certain governance variables. Of the 93 firms in the sample, 13 (or 14%) of them had CEO chairs, and 86% of them had separate roles for the two posts. A majority of the firms in the sample were run by indigenous CEOs, as only 37% of them had foreign chief executives. Further examination of the data showed a high degree of ownership concentration, with the largest shareholders owning on average 32.65% of equity. This compares with an average of 13.42% of shares owned by directors. Other governance variables examined at this stage were board size, for which an average of 8.45 was obtained, compared with the average of 6.29 for inside directors and 2.41 for outside directors. The data also revealed that directors on average own 13.42% of equity but this average masks a great deal of variation across the sample firms. Half of the firms reported director shareholding of less than 5% of equity; 4% of the firms in the sample reported average director shareholding of 62% of equity.

As will be seen in a subsequent section of this paper, an optimal size of ten was obtained for board size. Interestingly, an examination of the data further revealed that only 15.1% of the firms had the recommended board size, while 62 (or 66.7%) had board membership below and 17 (or 18.3%) above the optimum level.

A descriptive analysis of this sort may be helpful in offering some insight into the basic outlines of the underlying data upon which the analysis was based. By its nature, however, descriptive analysis has a major limitation - it does not lend itself to statistical

**tests and** consequently cannot be used to draw general conclusions about firms outside **the sample** or indeed about the same firms over a different time period.

# Regression results

 $\mathbf{X} J^{\text{e-start}}$  ky examining the effects of internal control mechanisms (director VV shareholding, board size, ownership concentration, outside directors and leverage) on firm performance. The results are presented in Table 2. Column 1 of the table shows the results obtained by regressing the four governance mechanisms on an important measure of firm performance, ROA. Both director shareholding and board size show no significant relationship with return on assets.

Tabfe 2: Coefficient estimates for Equation 1

|                         | Dependent variable    |                       |                        |                   |  |
|-------------------------|-----------------------|-----------------------|------------------------|-------------------|--|
|                         | ROA                   | ROE                   | Tobin-Q                | PE ratio          |  |
| Director shareholding   | -0.005968<br>(-1.232) | -0.026374<br>(-0.348) | -0.00001<br>(-1.062)   | -0.069<br>(-1.1)  |  |
| Board size              | 0.036345<br>(.577)    | -0.562231<br>(-0.571) | 0.003803<br>(5.700)"   | '0.423<br>(0.497) |  |
| Outside directors       | -0.097689<br>(-1.66)' | 0.947748<br>(1.029)   | -0.002724<br>(-3.95)'" | 0.14<br>(0.18)    |  |
| Ownership concentration | 0.005488<br>(1.459)   | 0.00387<br>0(0.066)   |                        | 0.011<br>(.215)   |  |
| $R^2$                   | 0.05                  | 0.01                  | 0.12                   | 0.02              |  |
| F                       | 2.33'                 | 0.36                  | 11.78"'                | 0.66              |  |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*).

A similar set of results (presented in the second column of the table for ROE and the fourth column for PE ratio) was obtained when the equation was estimated using ROE or PE ratio as the measure of firm performance. When ownership concentration was dropped and the equation estimated using the Tobin-Q as the measure of firm performance, the results differed from those for other measures. Column 3 shows that board size is significantly positively related to firm performance while the ratio of outside directors has the opposite effect.

These results show no discernible pattern and we are inclined to attribute this to a number of loopholes inherent in the specifications. Chief among the weaknesses is the failure to undertake a logarithmic transformation of the data before the application of

regression analysis. Logarithmic transformation has the advantage of dealing with the problems of heteroscedasticity. The exclusion of other variables is another. The results presented in Table 3 were obtained to address the first concern.

Table 3: Coefficient estimates for Equation 2

| Independent variables   | Dependent va | ariable   |                  |          |
|-------------------------|--------------|-----------|------------------|----------|
|                         | ROA          | ROE       | Tobin Q          | PE ratio |
| Director shareholding   | -0.058446    | -0.034103 | -0.19819         | -0.0642  |
|                         | (-1.064)     | (640)     | <b>(-5.1</b> r * | (-2.7)"' |
| Board size              | 0.044463     | 0.202393  | 0.028863         | 0.0429   |
|                         | (0.713)      | (3.35)''' | (0.634)          | (1.574)  |
| Outside directors       | -0.878889    | -0.263542 | -0.310642        | -0.0756  |
|                         | (-1.551)     | (-0.479)  | (-0.769)         | (-0.296) |
| Ownership concentration | 0.015664     | 0.01358   | 0.0159           | -0.0019  |
|                         | (2.348)"     | 9(2.100)" | (3.32)"'         | (-0.678) |
| $R^2$                   | 0.07         | 0.09      | 0.25             | 0.07     |
| F                       | 2.10"        | 3.89'"    | 12.69'"          | 2.86"    |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*).

As in Table 2, the results in Table 3 show that four alternative measures of firm performance were regressed against a set of four governance variables. In a number of ways, the logarithmic transformation has achieved some remarkable change in the results. For one, the F-statistic is significant at 5% or better. Secondly, ownership concentration has a significant positive effect in all but one case, PE ratio, where it is found not to be significant. This result does not support Adenikinju and Ayorinde (2001), who found no significant relationship between firm performance and ownership concentration. The conflicting results are perhaps due to the differences in the methods we use in measuring ownership concentration and in sample size taken. We took a sample size of 93 firms, while their sample numbered 73 firms.

Turning to individual coefficient estimates for each of the regressors, a clear pattern is observable. Director shareholding is significantly negatively related to firm performance in two of the four cases. This compares with outside directors and ownership concentration, which are not significant in all cases. This finding also does not support Adenikinju and Ayorinde (2001), who saw no significant relationship between firm performance and insider ownership. Again, the conflicting results are perhaps due to the differences in the methods we use in measuring ownership concentration and in sample size taken. In computing directors' shareholdings, we included only the shareholding of directors while they included those of directors and all other staff of the firms.

To address the issue of controlling for firm size, we included total assets as a control variable. The literature has advocated the use of total assets as a control variable (see, for example, Mayers et al. 1997; Sanders, 1998; Bhagat and Black, 2000). The results, as presented in Table 4, were obtained with total assets featuring as a control variable. In column 1, all the five variables with exception of outsider director are significant at the 5% level or better. In particular, director shareholding is negatively related to performance as measured by ROA. Similarly, board size is significant and positively related to firm performance as is ownership concentration. In column 3 it can be seen that director shareholding is significantly negatively related to Q. Both board size and ownership concentration exhibit a positive relationship to firm performance, but outside directors show no significant relationship. When the PE ratio was used as a measure of firm performance, the results (in column 4) show significant relationship to board size and a negative one to director shareholding.

Table 4: Coefficient estimates for Equation 3

| Independent variables   | Dependent variable |           |            |            |
|-------------------------|--------------------|-----------|------------|------------|
|                         | ROA                | ROE       | Tobin-Q    | PE ratio   |
| Director shareholding   | -0.115544          | -0.030733 | -0.258685  | -0.072490  |
|                         | (-2.223)**         | (-0.562)  | (-7.414)"" | (-2.997)"' |
| Board size              | 0.184279           | 0.194139  | 0.163717   | 0.060589   |
|                         | (2.907)"           | (2.913)'" | (3.767)'"  | (2.025)"   |
| Outside directors       | -0.760303          | -0.270543 | -0.237647  | -0.088727  |
|                         | (-1.449)           | (-0.490)  | (-0.681)   | (-0.348)   |
| Ownership concentration | 0.014893           | 0.013635  | 0.014714   | -0.002178  |
|                         | (2.412)"           | (2.100)"  | (3.546)"'  | (-0.760)   |
| Total assets            | -0.526620          | 0.031088  | -0.487051  | -0.065844  |
|                         | (-5.324)"'         | (0.299)   | (-7.264)'" | (-1.411)   |
| $R^2$                   | 0.21               | 0.09      | 0.44       | 0.09       |
| F                       | 8.44'"             | 3.11"'    | 24.16"'    | 2.70"      |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*).

#### Effects of rion-linearity

Extending the model in order to examine the effects of non-linearity as well as those of board independence, we modified the model in two ways. The first involved quadratic terms for board size and director shareholding, and the second involved two measures of board independence. The first measure of board independence is a dummy variable, taking a value of 0 for firms having a CEO Chair, and 1 otherwise. The results are presented in Table 5, divided into four columns, one each for alternative measures of firm performance.

Table 5: Coefficient estimates for Equation 4

| Independent variables | Dependent variable |           |            |            |  |
|-----------------------|--------------------|-----------|------------|------------|--|
|                       | ROA                | ROE       | Tobin-Q    | PE ratio   |  |
| Outside director      | -0.166984          | 0.484363  | -0.434273  | -0.258913  |  |
|                       | (-0.248)           | (0.692)   | (-1.017)   | (-0.813)   |  |
| Director shareholding | -0.116299          | -0.026241 | -0.241464  | -0.072582  |  |
|                       | (-2.259)"          | (-0.490)  | (-7.149)'" | (-2.976)"' |  |
| Director shareholding | -0.00781           | -0.005008 | -0.002968  | -0.006714  |  |
|                       | (-0.331)           | (-0.204)  | (-0.195)   | (-0.611)   |  |
| Board size            | 0.515857           | 0.597944  | 1.188012   | -0.142838  |  |
|                       | (1.006)            | (1.121)   | (3.383)"'  | (-0.553)   |  |
| CEO status            | 0.824726           | 1.04082   | -0.105101  | -0.279897  |  |
|                       | (1.692)*           | (2.052)** | (-0.341)   | (-1.271)   |  |
| Board size squares    | -0.019250          | -0.02339  | -0.057693  | 0.011829   |  |
|                       | (-0.669)           | (-0.781)  | (-2.949)"* | (0.825)    |  |
| Concentration         | 0.493277           | 0.52868   | 0.605746   | -0.028144  |  |
|                       | (2.586)'"          | (2.66)"'  | (4.886)"*  | (-0.318)   |  |
| Total assets          | -0.5473            | 0.0058    | -0.473662  | -0.058268  |  |
|                       | (-5.43)"*          | (0.056)   | (-7.151)"' | (-1.215)   |  |
| $R^2$                 | 0.23               | 0.13      | 0.5        | 0.11       |  |
| F                     | 5.84'"             | 2.95*"    | 18.26"'    | 2.12*"     |  |

Significant at 10% (\*); 5% (\*\*); 1% <\*\*\*).

A number of observations can be made concerning the results. First, in all cases, the F-statistic is significant at the 1% level, with R<sup>2</sup> varying from 0.11 for PE Ratio to 0.50 for Q. Second, one of the measures of board independence, outside directors, is not significant; the other measure, *CEOSTATUS*, is significant in two out of four cases and in both of those cases the dummy variable has a positive coefficient estimate, suggesting the need for separation of offices of CEO and Chair. Third, ownership concentration turned out to be significant in three out of four cases. It is striking that in each case the coefficient estimate is positive, implying that firms with concentrated ownership tend to perform better than those with diffused ownership. The results on Q, presented in column 3, require a close examination in view of certain peculiarities. Five out of eight variables are significant at the 1% level. In particular, both measures of board size are significant, with the quadratic one having a negative sign. In other words, the relationship between firm performance and board size is positive up to a point. Taking partial derivatives and solving for optimal values gave results suggesting an optimal value of ten for board size. Beyond this level a negative relationship is predicted to set in.

#### Managerial skills

Do firms in Nigeria with higher managerial skills and international exposure record better levels of performance than other firms in the country? On the assumption that a a foreign CEO would bring those attributes, we introduced a dummy variable taking a value of 1 for firms with expatriate CEOs, and 0 otherwise. The results are given in Table 6.

Table 6: Coefficient estimates for Equation 5

|                               | ROA       | ROE       | Tobin-Q    | PE ratio  |
|-------------------------------|-----------|-----------|------------|-----------|
| Outside director              | 0.67028   | 1.059945  | -0.118332  | -0.329075 |
|                               | (1.020)   | (1.517)   | (-0.260)   | (-0.984)  |
| Director shareholding         | -0.07368  | 0.027874  | -0.219761  | -0.049454 |
|                               | (-1.476)  | (0.525)   | (-6.165)"' | (-1.920)' |
| Director shareholding squares | -0.00459  | -0.004109 | -0.004920  | -0.01483  |
|                               | (-0.208)  | (-0.175)  | (-0.314)   | 5(-1.310) |
| Board size                    | -0.34837  | 0.132308  | 0.935705   | 0.180188  |
|                               | (-0.574)  | (0.205)   | (2.175)"   | (0.580)   |
| CEO foreign                   | 1.76286   | 1.707217  | 0.539628   | -0.03279  |
|                               | (6.453)"' | (5.879)"' | (2.783)'"  | (-0.229)  |
| CEO status                    | 1.53158   | 1.633460  | 0.154363   | -0.30036  |
|                               | (3.319)"' | (3.330)"' | (0.483)    | (-1.325)  |
| Board size sq                 | 0.03119   | 0.005379  | -0.043851  | -0.00729  |
|                               | (0.925)   | (0.150)   | (-1.828)'  | (-0.421)  |
| Concentration                 | -0,126776 | -0.104089 | 0.400093   | -0.05621  |
|                               | (-0.634)  | (-0.490)  | (2.812)"*  | (-0.549)  |
| Total assets                  | -0.477933 | 0.090546  | -0.445569  | -0.02821  |
|                               | (-5.09)"" | (0.908)   | (-6.583)'" | (-0.578)  |
| $R^2$                         | 0.39      | 0.28      | 0.5        | 0.07      |
| F                             | 10.52'"   | 6.44"'    | 15.62'"    | 1.13      |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*)

A number of observations can be made from the results in Table 6. We find that in three out of four cases, the coefficient estimate of the Foreign CEO dummy variable is positive and significant at the 1% level. This implies that firms with foreign CEOs tend to perform better than those with indigenous CEOs, a conclusion consistent with the works of Laing and Weir (1999) and Estrin et al. (2001), who stress the importance of managerial skills and business experience as a means of promoting firm performance. It

may also be that foreign CEOs would have internalized commonly accepted norms of international business practices, whereas indigenous CEOs would not have had the chance to do so. A closer look at the results also shows that despite the inclusion of an additional variable, the results bear a striking similarity to those presented in Table 5. Most of the coefficient estimates retained their signs and levels of significance. Further, the inclusion of the Foreign CEO dummy tended to improve the overall explanatory power of the model, with R<sup>2</sup> in most cases showing some improvements, as well as retaining its significance.

#### Ownership concentration: A double edged sword?

The issue of a non-linear relationship between firm performance and ownership concentration has been stressed in the literature. Is the implied stylized fact upheld by the data in Nigeria? This question was taken up by including a quadratic term for ownership concentration amongst the set of regressors; the results are presented in Table

We focus attention on one measure of ownership performance - Q - for a couple of reasons. Except for this measure of performance, the parameter estimates for the other measures are not stable; in fact, they tend to wander rather erratically. Moreover, the explanatory power of the model for each of the other measures of performance is well below that of the Q. From the results, seven of the nine parameter estimates are significant at 1 %, with the adjusted R<sup>2</sup> computed at 55.6%. We also observe a statistically significant negative relationship between director shareholding and firm performance, an unexpected finding considering the predicted positive relation between the two variables. The results also uphold the prediction that financial performance is better in firms run by foreign CEOs (who are assumed to have better international exposure and skills in modern management techniques). This conclusion seems to be in agreement with the findings of Laing and Weir (1999), who emphasize the importance of skills and entrepreneurship.

The finding that director shareholding is negatively related to firm performance is puzzling and one wonders what might have accounted for this unexpected result. There is the tendency for director shareholding to prevent takeover of the firm even when there is the need for this. We also take a cue from a newspaper (*The Punch*, 2003) report that some directors owed their firms hefty amounts and that a significant proportion of such amounts faced the prospects of turning into bad loans. If company shares were part of a portfolio of assets owed by directors to their firms, that might offer another piece of explanation for the puzzling finding. The omission of variables, especially sector dummy variables to capture the effects of variations across sectors, may be another reason for the unexpected results.

Table 7: Coefficient estimates for Equation 6

| Regressors                      |           |          |            |            |
|---------------------------------|-----------|----------|------------|------------|
|                                 | ROA       | ROE      | Tobin-Q    | PE ratio   |
| Outside directors               | -0.0951   | 0.723    | -0.469     | -0.142     |
|                                 | (-0.151)  | (1.167)  | (-1.208)   | (-0.459)   |
| Director shareholding           | -0.0887   | -0.01832 | -0.237     | -0.0717    |
|                                 | (-1.704)* | (-0.357) | (-7.27)*** | (-2.87)*** |
| Director shareholding squares   | -0.0103   | -0.00433 | 0.00154    | -0.006787  |
|                                 | (-0.441)  | (-0.189) | (0.106)    | (-0.603)   |
| Board size                      | 0.361     | 0.346    | 1.045      | -0.08453   |
|                                 | (0.676)   | (0.656)  | (3.120)*** | (-0.319)   |
| Board size squares              | -0.0063   | -0.005   | -0.04551   | 0.007448   |
|                                 | (-0.214)  | (-0.172) | (-2.438)** | (0.507)    |
| Expatriate CEOs                 | 1.436     | 1.444    | 0.572      | 0.06793    |
|                                 | (5.55)*** | (5.7)*** | (3.506)*** | (0.530)    |
| CEO status                      | -0.737    | -1.040   | 0.246      | 0.285      |
|                                 | (-1.633)  | (-2.3)** | (0.885)    | (1.352)    |
| Ownership concentration         | 0.791     | 2.32     | 4.355      | 2.037      |
|                                 | (0.428)   | (1.27)   | (3.680)*** | (2.268)**  |
| Ownership concentration squares | -0.123    | -0.36    | -0.606     | -0.325     |
|                                 | (-0.431)  | (-1.27)  | (-3.33)*** | (-2.348)** |
| Total assets                    | -0.527    | -0.0227  | -0.556     | -0.0646    |
|                                 | (-5.3)*** | (-0.232) | (-8.96)*** | (-1.350)   |
| $R^2$                           | 0.346     | 0.261    | 0.553      | 0.042      |
| F                               | 9.401***  | 6.62***  | 19.811***  | 1.636      |

Significant at 10% f); 5% (\*\*); 1% (\*\*\*)

## Accounting for industry variations

The relationship between firm performance and governance mechanisms might well vary from one sector of the exchange to another. To address this issue, 13 dummy variables were included to account for possible variations in the relationship because of the differences in risk exposures of firms operating in the 14 different sectors from which the sample was drawn. The results are presented in Table 8.

Table 8: Coefficient estimates for Equation 7

| Regressors                      | Dependent variable  |                    |                   |                                  |
|---------------------------------|---------------------|--------------------|-------------------|----------------------------------|
|                                 | ROA                 | ROE                | Tobin-Q           | PE ratio                         |
| Outside directors               | -1.064              | -0.664             | -1.44             | 0.226                            |
|                                 | (-1.306)            | (-0.810)           | (-3.26)***        | (0.544)                          |
| Director shareholding           | -0.06539            | -0.0302            | -0.191            | -0.07715                         |
|                                 | (-0.896)            | (-0.411)           | (-4.78)***        | (-2.1)**                         |
| Director shareholding squares   | 0.05045             | 0.0304             | 0.03915           | -0.0029                          |
|                                 | (1.597)             | (0.957)            | (2.25)**          | (-0.179)                         |
| Board sìze                      | -0.183              | -0.499             | 1.330             | 0.252                            |
|                                 | (-0.264)            | (-0.71)            | (3.23)***         | (0.63)                           |
| Board size squares              | 0.0208              | 0.0418             | -0.065            | -0.011                           |
|                                 | (0.555)             | (1.106)            | (-2.92)***        | (-0.5)                           |
| Expatriate CEO                  | 1.345               | 1.315              | 0.251             | 0.213                            |
|                                 | (3.9)***            | (3.4)***           | (1.16)            | (1.04)                           |
| CEO status                      | -0.983              | -0.654             | 0.629             | -0.326                           |
|                                 | (-1.47)             | (-0.97)            | 0.629(1.74)       | (-0.96)                          |
| Ownership concentration         | -0.005              | 1,504              | 3.791             | 2.56                             |
|                                 | (-0.003)            | (0.731)            | (3.32)***         | 5(2.39)**                        |
| Ownership concentration squares | 0.0494              | -0.229             | -0.547            | -0.42                            |
|                                 | (0.16)              | (-0.73)            | (-3.18)***        | (-2.6)**                         |
| Total assets                    | -0.73               | -0.00516           | -0.585            | -0.142                           |
|                                 | (-4.6)***           | (-0.032)           | (-6.68)***        | (-1.71)*                         |
| Banking                         | -1.41               | -1.447             | -0.779            | 0.628                            |
|                                 | (-1.87)*            | (-1.92)*           | (-1.85)*          | (1.56)                           |
| Breweries                       | -1.49               | -1.711             | 0.135             | 0.436                            |
|                                 | (-1.4)              | (-1.62)            | (0.2)             | (0.82)                           |
| Building                        | -2.15               | -1.800             | -1.077            | 0.248                            |
|                                 | (-2.6)***           | (-2.2)**           | (-2.43)**         | (0.610)                          |
| Conglomerates                   | -2.342              | -2.079             | -2.22             | 0.292                            |
|                                 | (-2.59)**           | (-2.3)**           | (-4.6)***         | (0.67)                           |
| Construction                    | -2.337              | -1.233             | -1.185            | -0.104                           |
|                                 | (-2.56)**           | (-1.3)             | (-2.4)**          | (-0.22)                          |
| Food and beverages              | -1.054              | -0.22              | -0.033            | 0.230                            |
|                                 | (-1.338)            | (-0.3)             | (-0.078)          | (0.59)                           |
| Health                          | -2.405              | -1.461             | -0.835            | 0.659                            |
|                                 | (-2.13)**           | (-1.29)            | (-1.357)          | (1.11)                           |
| Industrial                      | -3.544<br>(-3.6)*** | -1.993<br>(-2.0)** | 0.0612<br>(0.112) | 0.653<br>(1.29)<br><i>Contin</i> |

Table 8, continued\_

| Regressors | Dependen  | t variable |            |          |
|------------|-----------|------------|------------|----------|
|            | ROA       | ROE        | Tobin-Q    | PE ratio |
| Insurance  | -1.632    | -0.635     | -1.35      | 0.0784   |
|            | (-2.32)** | (-0.9)     | (-3.48)*** | (0.22)   |
| Packaging  | -3.976    | -2.19      | -1.669     | 0.376    |
|            | (-4.4)*** | (-2.4)**   | (-3.43)*** | (0.81)   |
| Petroleum  | -0.816    | 0.168      | 0.589      | 0.0874   |
|            | (-0.92)   | (0.19)     | (1.24)     | (0.20)   |
| Textiles   | -3.152    | -1.695     | -1.432     | -0.58    |
|            | >3.1)***  | (-1.64)    | (-2.54)**  | 6(-1.03) |
| $R^2$      | 0.48      | 0.379      | 0.741      | 0.099    |
| F          | 7.161***  | 5,083***   | 19.2***    | 1.663**  |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*).

In three important ways, the inclusion of sector dummies offered further insights into the nature of the relationship between governance variables and firm financial performance. First, the automobile sector showed a better level of performance than the textile, conglomerate, insurance, construction and packaging sectors of the exchange A second result is that despite the extension of the model, the nature of the relationship between board size and firm performance has remained unchanged, with the results predicting an optimal size of ten board members.

A more interesting insight offered by the inclusion of sector dummies in the regression analysis concerns the relationship between firm performance and governance variables notably ownership concentration and director shareholding. As in the previous results a statistically significant relationship is found between firm performance and the two governance variables. Given the negative coefficient estimate of the quadratic term for the concentration variable, performance is predicted to rise within a certain range and fall thereafter. Given the positive sign of the coefficient estimate for the quadratic term for director shareholding, it is predicted that beyond a certain level of director shareholding, further ownership of shares by directors would lead to improvements in performance This would seem rather perplexing, for the literature suggests a limit within which such a positive relationship can be expected to hold.

Do the results therefore run counter to theoretical expectation? To answer this question we refer to the coefficient estimates of the two quadratic terms in the model. A negative coefficient estimate for the quadratic term for ownership concentration implies an inverted U- shape for the relationship between concentration and firm performance. Taking partial derivatives and solving for optimal values we obtained results implying that beyond an ownership concentration of 32.46%, a negative relationship will set in. By the same

token, a positive coefficient estimate for the quadratic term for director shareholding implies a U-shaped relationship between director shareholding and firm performance. Taking partial derivatives and solving for optimal values we obtained results indicating that beyond director shareholding of 8.94%, a positive relationship is predicted between firm performance and director shareholding. In view of this, we propose that there is a limit to which this relationship might hold, although the U-shaped nature of the function suggests otherwise. As directors own more and more shares, this will increase ownership concentration. If the level of director shareholding continued to rise and thereby caused the level of ownership concentration to rise beyond the threshold of 32.46%, would the relationship between director shareholding still be positive in view of the U-shaped nature of the function?

The answer depends on whether performance is falling (owing to concentration effects) faster than it is rising (owing to director shareholding effect). The coefficient estimates of the two quadratic terms indicate that the absolute value for concentration is higher than that of director shareholding. Thus, after ownership concentration of 32.46%, the negative effects will outweigh the positive effect of director shareholding. Hence the negative effects of concentration seem to prevent director shareholding from having an unlimited range within which to exhibit a positive correlation with performance. These results are tentative and further investigation is required to address these and related issues. Such issues include, for example, the need to estimate the level of director shareholding required to raise the level of ownership concentration to the threshold level.

The results brought about major, unexpected changes in the coefficient estimate. Director shareholding is significant (at 10%), although exhibiting a linear relationship; *CEO STATUS* has a negative and significant coefficient estimate; outside directors shows a significant and positive relationship with performance. Interestingly, debt turns out to be significant and positively associated with performance. This is expected as supported by the literature. An R<sup>2</sup> value of 92% is very worrying for this sort of data, however, and we are inclined to believe that the book value of debt used to compute the revised Q could be the source of these unexpected results. The results therefore strengthen the case for further research.

# Effects of leverage

The regression analysis was also extended to incorporate two new elements. The first was the need to consider leverage in the computation of the Q and the second was to include debt as a control variable. The results are presented in Table 9.

#### Table 9: Coefficient estimates for Equation 8

| Outside directors     | 2.052<br>(3.116)***               |
|-----------------------|-----------------------------------|
| Director shareholding | <br>0.177<br>0 - <sup>827</sup> T |

| Table 9, Continued from         |                       |  |
|---------------------------------|-----------------------|--|
| Director shareholding squares   | -0.02105<br>(-0.668)  |  |
| Board size                      | -2.074<br>(-3.023)*** |  |
| Board size squares              | 0.121<br>(3.122)***   |  |
| Expatriate CEO                  | 3.125<br>(4.686)***   |  |
| CEO status                      | -2.017<br>(-2.659)*** |  |
| Ownership concentration         | 4.817<br>(2.514)**    |  |
| Ownership concentration squares | -1.028<br>(-3.712)*** |  |
| Total assets                    | -1.292<br>(-9.318)*** |  |
| Debt                            | 0.446<br>(7.176)***   |  |
| Banking                         | 1.958<br>(2.951)***   |  |
| Breweries                       | -1.144<br>(-1.345)    |  |
| Building                        | 5.023<br>(5.664)***   |  |
| Conglomerates                   | 3.015<br>(3.784)***   |  |
| Construction                    | -0.902<br>(-1.164)    |  |
| Industrial                      | 1.107<br>(1.836)*     |  |
| Insurance                       | 0.05269<br>(0.128)    |  |
| Packaging                       | 1.498<br>(2.465)**    |  |
| Petroleum                       | -0.134<br>(-0.204)    |  |
| Textiles                        | 1.255<br>(0.783)      |  |
| R <sup>2</sup>                  | 0.919                 |  |
| F                               | 46.533***             |  |

Significant at 10% (\*); 5% (\*\*); 1% (\*\*\*).

## 8. Concluding remarks

here has been a renewed interest within academic circles as well as amongst policy makers in both government and industry in the need to strengthen mechanisms to ensure that managers and directors take measures to protect the interest of a firm's stakeholders. The events at Enron and other cases of spectacular failure have helped to bring to the limelight the important role that the strengthening of governance mechanisms could play to improve firm performance. This study uses pooled OLS regression analysis on panel data for the period 1996 through 1999 for a sample of 93 firms listed on the Nigerian Stock Exchange to examine the relationship between internal governance mechanisms and firm financial performance.

Apart from supporting the view that separating the posts of CEO and Chair works in favour of the firm, our results support the need to maintain a board size of ten persons, in line with findings from other countries. The results have the implication that regulatory agencies should encourage firms that have not already done so to separate the two posts. There is also need to encourage firms to achieve a reasonable board size since overly large boards may be detrimental to the firm.

The results of the study point to the need for a reasonable number of individuals and/ or corporate bodies with more than a typical share of equity of the firm as this will encourage them to undertake the monitoring process. Unlike the findings in developed countries, our results show no significant evidence to support the idea that outside directors help promote firm performance. This suggests the need for the regulatory authorities to reassess the procedures for the appointment of outside directors in order to remove the influence of CEOs from the appointment process.

Another feature of our results is the finding that firms run by expatriate CEOs tend to perform better than those run by indigenous ones. We are inclined to attribute this finding to the tendency for foreign CEOs to have better managerial and administrative skills, and this has the implication for the need to pursue policies to improve the managerial and administrative skills of indigenous CEOs in a bid to bring them into parity with their foreign counterparts.

Our results also show that leverage has significant positive influence on firm performance, indicating the tendency for firms with higher levels of debt as a proportion of equity to perform better, a finding that is consistent with the literature.

On the whole, the results of this study come out best for a capital market based measure of performance (Q). This is understandable because, unlike the capital market based measure, other measures of performance are subject to accounting artefacts and do not account for risk differences.

These results are by no means conclusive for a couple of reasons. First the sample itself was determined by data availability, not by a probability criterion. A second limitation concerns the method of statistical analysis, which relied on the standard OLS regression rather than on the more robust, quintile regressions or even panel data analysis. These limitations should be borne in mind in any attempt to undertake research in this area on the Nigerian Stock Exchange.

## **Note**

1. If Y = firm performance, X = board size and  $X^2$  = board size squared, and taking the coefficients of X and  $X^2$  under Q specification, then: Y = 1.188012X + (-0.057693 $X^2$ ). \ dy/dx = 1.188012 + 2(-0.057693X) = 1.188012 - 0.115386X. Now solving for optimal value: 1.188012 - 0.115386X = 0, \ -0.115386X = -1.188012, then,  $X = -U88012/-0.115386 = 10.30^8$  10.

### References

- Adenikinju, O. and F. Ayorinde. 2001. "Ownership structure, corporate governance and corporate performance: The case of Nigerian quoted companies". Unpublished Final Report presented at the AERC biannual research workshop, Nairobi, May.
- Agrawal, A. and C.R. Knoeber. 1996. "Firm performance and mechanisms to control agency problems between managers and shareholders". *Journal of Financial and Quantitative Analysis*, 31(3, September): 377-97.
- Bhagat, S. and B. Black. 1999. "The non-correlation between board independence and long-term firm performance". *Journal of Corporation Law*, 27: 231-71.
- Bhagat, S. and B. Black. 2000. "Board independence and long-term firm performance". Stanford Law School Working Paper No. 188.
- Bhagat, S. and B. Black. 1999. "The uncertain relationship between board composition and firm performance". *Business Lawyer*, 54: 921-63.
- Dean, J. W. and T. Audreyeva. 2001. "Privatisation, ownership structure and company performance: The case of Ukraine". *Journal for International Innovation, Development and Transition*, Institute of Macroeconomic Analysis and Development (IMAD), Ljubijana, Slovenia, vol. 5: 62-72.
- DeAngelo, H. and L. DeAngelo. 1985. "Managerial ownership of voting rights: A study of public corporations with dual classes of common stock". *Journal of Financial Economics*, 14: 33-69.
- Demirguc-Kunt, A. 1992. "Developing country capital structures and emerging stock markets". *World Bank Policy Research Working Papers*, WPS 933, pp. 1-40, July.
- Demirguc-Kunt, A. and R. Levine. 1996. "Stock market development and financial intermediaries: Stylized facts". *The World Bank Economic Review*, 10(2): 231.
- Demsetz, H. and K. Lehn. 1985. "The structure of corporate ownership: Causes and Consequences". *Journal of Political Economy*, 93(6): 1155-77.
- Emenuga, C. 1998. "The Nigerian capital market and Nigeria's economic performance". Proceedings of the One-Day Seminar of the Nigerian Economic Society, Ibadan, Nigeria, 21 January.
- Estrin, S., J. Konings, Z. Zolkiewski and M. Angelucci. 2001. "The effect of ownership and competitive pressure on firm performance in transition countries: Micro evidence from Bulgaria, Romania and Poland". Paper presented at LICOS, K.U. Leuven; a Phare Ace workshop on Competition Policy in Romania, 7 August.
- Fama, E.F. 1980. "Agency problems and the theory of the firm". *Journal of Political Economy*, 88(2, April): 288-307.
- Fosberg, R. 1989. "Outside directors and managerial monitoring". *Akron Business and Economic Review*, 20: 24-32.

Garcia, V.F. and L. Liu. 1999. "Macroeconomic determinants of stock market development". *Journal of Applied Economics*, 11(1, May): 29-59.

- Genmax. 1998. Leading Companies in Industry and. Commerce. Lagos: Genmax Communications Nigeria Ltd.
- Gorton, G. and F. Schmid. 1996. "Universal banking and performance of German firms". Working Paper 5453, National Bureau of Economic Research, Cambridge, Massachusetts USA.
- Grosfeld, I. 2002. "Exploring the link between privatization and other policies in transition". Paper presented at Beyond Transition conference, organized by Centre for Anlaysis of Social Exclusion, London School of Economics, Falenty, 12-13 April.
- Gupta, V. 1999. SPSS for Beginners. USA: JV Books, Inc.
- Hermalin, B.E. and M.S. Weisbach. 1991. "The effects of board composition and direct incentives on firm performance". *Financial Management*, Winter: 101-12.
- Holderness, C. and D. Sheehan. 1988. "The role of majority shareholders in publicly held corporations". *Journal of Financial Economics*, 20: 317-46.
- Jensen, Michael C. 2001. "Value maximization, stakeholder theory, and the corporate objective function". Working Paper No. 01-01, Harvard Business School.
- Jensen, M.C. and W.H. Meckling. 1976. "Theory of the firm: Managerial behaviour, agency costs and ownership structure". *Journal of Financial Economics*, 2: 305-60.
- John, K. and L.W. Senbet .1998. "Corporate governance and board effectiveness" *Journal of Banking and Finance*, 22: 371-403.
- Kang, J. and A. Shivdasani. 1995. "Firm performance, corporate governance and top executive turnover in Japan". *Journal of Financial Economics*, 38: 29-58.
- Kaplan, S. and B. Minton. 1994. "Appointment of outsiders to Japanese boards: Determinants and implications for managers". *Journal of Financial Economics*, 36: 225-57.
- Klapper, L.F. and I. Love. 2002. "Corporate governance, investor protection, and performance in emerging markets". *World Bank Policy Research Working Paper* 2818, April.
- Klein, A. 1998. "Firm performance and board committee structure". *Journal of Law and Economics*, 41: 275-303.
- Laing, D. and C.M. Weir. 1999. "Governance structures, size and corporate performance in UK firms". *Management Decision*, 37(5): 457-64.
- Lipton, M. and J.W. Lorsch. 1992. "A Modest Proposal for improved corporate governance". *Business Lawyer*, 48(1) 59-77.
- Loderer, C. and K. Martin. 1997. "Executive stock ownership and performance: Tracking faint traces", *Journal of Financial Economics*, 45: 223-55,
- Mayers, D., A. Shivdasani and C.W. Smith. 1997. "Board composition and shareholders wealth: The case of management buyout". *Financial Management*, 21: 58-72.
- McConnell, J.J. and H. Servaes. 1990. "Additional evidence on equity ownership and corporate value". *Journal of Financial Economics*, 27(2 October): 595-613.

- Mehran, H. 1995. "Executive compensation structure, ownership and firm performance". *Journal of Financial Economics*, 38: 163-84.
- Metrick, A. and J. Ishii. 2002. "Firm-level corporate governance". Paper presented at Global Corporate Governance Forum Research Network Meeting, Washington, D.C. April.
- Monks, R.A.G. and N. Minow. 1995. "Corporate governance on equity ownership and corporate value". *Journal of Financial Economics*, 20: 293-315.
- Morck, R., A. Schleifer and R.W. Vishny. 1988. "Management ownership and market valuation: An empirical analysis". *Journal of Financial Economics*, 20: 293-315.
- Nigerian Stock Exchange. 2000. Fact Book. Lagos.
- Nor, F.M., R.M. Said and H. Redzuan. 1999. "Structure of ownership and corporate financial performance: A Malaysian case". *Malaysian Management Review*, December: 44-8.
- Oxelheim, L. and T. Randoy. 2001. "The impact of foreign board membership on firm value". *Working Paper No. 567*. The Research Institute of Industrial Economics (IUI), Stockholm, Sweden.
- Pallant, J. 2004. SPSS Survival Manual: A Step by Step Guide to Data Analysis Using SPSS. Maidenhead, Berkshire, United Kingdom: Open University Press, McGraw-Hill Education.
- Pinteris, G. 2002. "Ownership Structure, Board Characteristics and Performance of Argentine Banks". Mimeo. Department of Economics, University of Illinois.
- *Punch, The* (Nigerian daily newspaper). "Directors owe banks N29 billion CBN". 31 March, p. 22.
- Renneboog, L. 2000. "Ownership, managerial control and governance of companies listed on the Brussels stock exchange". *Journal of Banking and Finance*, 24(12, December): 1959-95.
- Ross, S. 1973. "The economic theory of agency: The principal's problem". *American Economic Review*, 63(2): 134-39.
- Sanders, W.M.G. 1998. "Internationalisation and firm governance: The role of CEO compensation, top team composition, and board structure". *Academy of Management Journal*, 41(2): 158-78.
- Shleifer, A. and R.W. Vishny, 1997. "A survey of corporate governance". *Journal of Financial Economics*, 52(2): 737-83.
- Weisbach, M. 1988. "Outside directors and CEO turnover". *Journal of Financial Economics*, 20: 431-60.
- Yeboah-Duah, K. 1993. "Stock ownership and the performance of the firm in Malaysia". *Capital Market Review*, 1(2): 83-108.
- Yermack, D. 1996. "Higher market valuation of companies with a small board of directors". *Journal of Financial Economics*, 40: 185-211.

# Appendix: List of companies used in the sample

Aba Textiles Pic

Aboseldehyde Laboratories pic

Afprint Nigeria Pic African Petroleum Pic A.G. Leventis (Nigeria) Pic

Agip (Nigeria) Pic AIICO Insurance pic

Aluminium Extrusion Industries

Aiumaco Pic

Amicable Assurance Pic

Arbico PLC

Ashaka Cement Pic

Avon Crowncaps & Containers Pic Benue Cement Company Pic

**BCN Pic** 

BOC Gases Pic Cappa & D'Alberto Pic Carnaudmetalbox Pic

Cement Company of Northern Nigeria Pic

Chartered Bank Pic

Cornerstone Insurance Pic Costain (West Africa) Plc Crusader Insurance Plc

Delta Glass Plc Dumez Plc

Dunlop Nigeria Plc Ekocorp Plc

Eko International Bank Plc

Enpee Plc

Evans Medical Plc

First Aluminium Nigeria Plc First Bank of Nigeria Pic Flour Mills of Nigeria Pic FSB International Bank Plc

G. Cappa Plc

Golden Guinea Breweries Plc Grommac Industries Plc Guaranty Trust Bank Plc Guinea Insurance Plc
Guinness Nigeria Plc
Hallmark Bank Plc
Inland Bank Plc
Intra Motors Plc
John Holt Plc
Julius Berger Plc

LASACO Insurance Plc Law Union & Rock Insurance Plc

Lever Brothers Nigeria Plc

Liberty Bank Plc

Lion Bank of Nigeria Plc May & Baker Nigeria Pic Mobil Oil Nigeria Plc Morison Industries Pic NAL Merchant Bank Plc

National Oil & Chemical Marketing

Company Plc NCR (Nigeria) Plc

Neimeth Intl Pharmaceuticals Plc

Nem Insurance Plc

Nestle Foods Nigerian Plc

Niger Insurance Plc

Nigerian Bottling Company Plc Nigerian Breweries Plc

Nigerian Wire & Cable Plc Nigerian Wire Industries Plc

Nivamco Pic

Northern Nigeria Flour Mills Plc Paterson Zochonis Industries Plc

PharmaDeko Plc

Poly Products Nigeria Plc Prestige Assurance Plc P.S. Mandrides Plc

Royal Exchange Assurance Plc R.T. Briscoe Nigeria Pic

Seven-up Bottling Company Plc

SmithklineBeecham Plc Sun Insurance Plc Texaco Nigeria Plc Total Nigeria Plc Towergate Insurance Plc Trans International Bank Plc Tripple Gee & Company Pic **UAC** of Nigeria Plc Unic Insurance Plc Union Dicon Salt Plc Unipetrol Nigeria Plc United Nigeria Textile Mills Plc Universal Trust Bank Plc Van Leer Containers Plc Vitafoam Nigeria Plc Wema Bank Plc West African Portland Cement Pic WAPIC Insurance Plc WTN Plc

### Other recent publications in the AERC Research Papers Series:

The Behaviour of Income Velocity in Tanzania 1967—1994, by Michael O.A. Ndanshau, Research Paper 50.

Consequences and Limitations of Recent Fiscal Policy in Cote d'Ivoire, by Kouassy Oussou and Bohoun Bouabre, Research Paper 51,

Effects of Inflation on Ivorian Fiscal Variables: An Econometric Investigation, by Eugene Kouassi, Research Paper 52.

European Economic Integration and the Franc Zone: The Future of the CFA Franc after 1999, Part II, by Allechi M'Bet and Niamkey A. Madeleine, Research Paper 53.

Exchange Rate Policy and Economic Reform in Ethiopia, by Asmerom Kidane, Research Paper 54.

The Nigerian Foreign Exchange Market: Possibilities for Convergence in Exchange Rates, by P. Kassey Garba, Research Paper 55.

Mobilizing Domestic Resources for Economic Development in Nigeria: The Role of the Capital Market, by Fidelis O. Ogwumike and Davidson A. Omole, Research Paper 56.

Policy Modelling in Agriculture: Testing the Response of Agriculture to Adjustment Policies in Nigeria, by Mike Kwanashie, Abdul-Ganiyu Garba and Isaac Ajilima, Research Paper 57.

Price and Exchange Rate Dynamics in Kenya: An Empirical Investigation (1970—1993), by Njuguna S. Ndung'u, Research Paper 58.

Exchange Rate Policy and Inflation: The Case of Uganda, by Barbara Mbire, Research Paper 59.

Institutional, Traditional and Asset Pricing Characteristics of African Emerging Capital Markets, by Ino L. Inanga and Chidozie Emenuga, Research Paper 60.

Foreign Aid and Economic Performance in Tanzania, by Timothy S. Nyoni, Research Paper 61,

Public Spending, Taxation and Deficits: What Is the Tanzanian Evidence? by Nehemiah Osoro, Research Paper 62.

Adjustment Programmes and Agricultural Incentives in Sudan: A Comparative Study, by Nasredin A. Hag Elamin and Elsheikh M. El Mak, Research Paper 63.

Intra-industry Trade between Members of the PTA/COMESA Regional Trading Arrangement, by Flora Mndeme Musonda, Research Paper 64.

Fiscal Operations, Money Supply and Inflation in Tanzania, by A.A.L. Kilindo, Research Paper 65.

Growth and Foreign Debt: The Ugandan Experience, by Barbara Mbire, Research Paper 66.

Productivity of the Nigerian Tax System: 1970-1990, by Ademola Aiiyo, Research Paper 67.

Potentials for Diversifying Nigeria's Non-Oil Exports to Non-Traditional Markets, by A. Osuntogun, C.G. Edordu and B.O. Oramah, Research Paper 68,

Empirical Studies of Nigeria's Foreign Exchange Parallel Market II: Speculative Efficiency and Noisy Trading, by Melvin Ayogu, Research Paper 69.

Effects of Budget Deficits on the Current Account Balance in Nigeria: A Simulation Exercise, by Festus O. Egwaikhide, Research Paper 70.

Bank Performance and Supervision in Nigeria: Analysing the Transition to a Deregulated Economy, by O.O. Sobodu and P.O. AM ode, Research Paper 71.

Financial. Sector Reforms and Interest Rate Liberalization: The Kenya Experience, by R.W. Ngugi and J.W. Kabubo. Research Paper 72.

Local Government Fiscal Operations in Nigeria, by Akpan H. Ekpo and John E.U. Ndebbio, Research Paper 73.

Tax Reform and Revenue Productivity in Ghana, by Newman Kwadwo Kusi, Research Paper 74.

Fiscal and Monetary Burden of Tanzania's Corporate Bodies: The Case of Public Enterprises, by H.P.B. Moshi. Research Paper 75.

Analysis of Factors Affecting the Development of an Emerging Capital Market: The Case of the Ghana Stock Market, by Kofi A. Osei, Research Paper 76.

Ghana: Monetary Targeting and Economic Development, by Cletus K. Dordunoo and Alex Donkor, Research Paper 77.

The Nigerian Economy: Response of Agriculture to Adjustment Policies, by Mike Kwanashie, Isaac Ajilima and Abdul-Ganiyu Garba, Research Paper 78.

Agricultural Credit under Economic Liberalization and Islamization in Sudan, by Adam B. Elhiraika and Sayed A. Ahmed, Research Paper 79.

Study of Data Collection Procedures, by Ademola Ariyo and Adebisi Adeniran, Research Paper 80. Tax Reform and Tax Yield in Malawi, by C. Chipeta, Research Paper 81.

Real Exchange Rate Movements and Export Growth: Nigeria, 1960-1990, by Oluremi Ogun, Research Paper 82.

Macroeconomic Implications of Demographic Changes in Kenya, by Gabriel N. Kirori and Jamshed Ali, Research Paper 83.

An Empirical Evaluation of Trade Potential in the Economic Community of West African States, by E. Olawale Ogunkola, Research Paper 84.

Cameroon's Fiscal Policy and Economic Growth, by Aloysius Ajab Amin, Research Paper 85.

Economic Liberalization and Privatization of Agricultural Marketing and Input Supply in Tanzania: A Case Study of Cashewnuts, by Ngila Mwase, Research Paper 86.

Price, Exchange Rate Volatility and Nigeria's Agricultural Trade Flows: A Dynamic Analysis, by A.A. Adubi and F. Okunmadewa, Research Paper 87.

The Impact of Interest Rate Liberalization on the Corporate Financing Strategies of Quoted Companies in Nigeria, by Davidson A. Omole and Gabriel O. Falokun, Research Paper 88.

The Impact of Government Policy on Macroeconomic Variables, by H.P.B. Moshi and A.A.L. Kilindo, Research Paper 89.

External Debt and Economic Growth in Sub-Saharan African Countries: An Econometric Study, by Milton A. Ivoha, Research Paper 90.

Determinants of Imports in Nigeria: A Dynamic Specification, by Festus O. Egwaikhide, Research Paper 91.

Macroeconomic Effects of VAT in Nigeria: A Computable General Equilibrium Analysis, by D. Olu Ajakaiye, Research Paper 92.

Exchange Rate Policy and Price Determination in Botswana, by Jacob K. Atta, Keith R. Jefferis, Ita Mannathoko and Pelani Siwawa-Ndai, Research Paper 93.

Monetary and Exchange Rate Policy in Kenya, by Njuguna S. Ndung'u, Research Paper 94.

Health Seeking Behaviour in the Reform Process for Rural Households: The Case of Mwea Division, Kirinyaga District, Kenya, by Rose Ngugi, Research Paper 95.

Trade Liberalization and Economic Performance of Cameroon and Gabon, by Ernest Bamou, Research Paper 97.

Quality Jobs or Mass Employment, by Kwabia Boateng, Research Paper 98.

Real Exchange Rate Price and Agricultural Supply Response in Ethiopia: The Case of Perennial Crops, by Asmerom Kidane, Research Paper 99.

Determinants of Private Investment Behaviour in Ghana, by Yaw Asante, Research Paper 100.

An Analysis of the Implementation and Stability of Nigerian Agricultural Policies, 1970-1993, by P. Kassey Garba, Research Paper 101.

Poverty, Growth and Inequality in Nigeria: A Case Study, by Ben E. Aigbokhan, Research Paper 102. Effect of Export Earnings Fluctuations on Capital Formation, by Godwin Akpokodje, Research Paper 103.

Nigeria: Towards an Optimal Macroeconomic Management of Public Capital, by Melvin D. Ayogu, Research Paper 104.

International Stock Market Linkages in South Africa, by K.R, Jefferis, C.C. Okeahalam and T.T. Matome, Research Paper 105.

An Empirical Analysis of Interest Rate Spread in Kenya, by Rose W, Ngugi, Research Paper 106

The Parallel Foreign Exchange Market and Macroeconomic Perfromance in Ethiopia, by Derrese
Degefa, Reseach Paper 107.

Market Structure, Liberalization and Performance in the Malawi Banking Industry, by Ephraim W. Chirwa, Research Paper 108.

Liberalization of the Foreign Exchange Market in Kenya and the Short-Term Capital Flows Problem, by Njuguna S. Ndung'u, Research Paper 109.

External Aid Inflows and the Real Exchange Rate in Ghana, by Hairy A. Sackey, Research Paper 110.

Formal and Informal Intitutions' Lending Policies and Access to Credit by Small-Scale Enterprises in Kenya: An Empirical Assessment, by Rosemary Atieno, Research Paper 111.

- Financial Sector Reform, Macroeconomic Instability and the Order of Economic Liberalization: The Evidence from Nigeria, by Sylvanus I. Ikhinda and Abayomi A. Alawode, Research Paper 112.
- The Second Economy and Tax Yield in Malawi, by C, Chipeta, Research Paper 113.
- Promoting Export Diversification in Cameroon: Toward Which Products? by Lydie T, Bamou, Research Paper 114.
- Asset Pricing and Information Efficiency of the Ghana Stock Market, by Kofi A. Osei, Research Paper 115.
- An Examination of the Sources of Economic Growth in Cameroon, by Aloysius Ajab Amin, Research Paper 116.
- Trade Liberalization and Technology Acquisition in the Manufacturing Sector: Evidence from Nigeria, by Ayonrinde Folasade, Research Paper 117.
- Total Factor Productivity in Kenya: The Links with Trade Policy, by Joseph Onjala, Research Paper 118. Kenya Airways: A Case Study of Privatization, by Samuel Oyieke, Research Paper 119,
- Determinants of Agricultural Exports: The Case of Cameroon, by Daniel Gbetnkon and Sunday A. Khan, Research Paper 120.
- Determinants of Regional Poverty in Uganda, by Francis Okurut, Jonathan Odwee and Asaf Adebua, Research Paper 122
- Exchange Rate Policy and the Parallel Market for Foreign Currency in Burundi, by Janvier D. Nkurunziza, Research Paper 123.
- Structural Adjustment, Poverty and Economic Growth: An Analysis for Kenya, by Jane Kabubo-Mariara and Tabitha W. Kiriti, Research Paper 124.
- Liberalization and Implicit Government Finances in Sierra Leone, by Victor A.B. Davis, Research Paper 125.
- Productivity, Market Structure and Trade Liberalization in Nigeria, by Adeola F. Adenikinju and Louis N. Chete, Research Paper 126.
- Productivity Growth in Nigerian Manufacturing and Its Correlation to Trade Policy Regimes/Indexes (1962-1985), by Louis N. Chete and Adeola F. Adenikinju, Research Paper 127.
- Financial Liberalization and Its Implications for the Domestic Financial System: The Case of Uganda, by Louis A. Kasekende and Michael Atingi-Ego, Research Paper 128,
- Public Enterprise Reform in Nigeria: Evidence from the Telecommunications Industry, by Afeikhena Jerome, Research Paper 129.
- Food Security and Child Nutrition Status among Urban Poor Households in Uganda: Implications for Poverty Alleviation, by Sarah Nakabo-Sswanyana, Research Paper 130.
- Tax Reforms and Revenue Mobilization in Kenya, by Moses Kinyanjui Muriithi and Eliud Dismas Moyi, Research Paper 131.
- Wage Determination and the Gender Wage Gap in Kenya: Any Evidence of Gender Discrimination? by. Jane Kabubo-Mariara, Research Paper 132.
- Trade Reform and Efficiency in Cameroon's Manufacturing Industries, by Ousmanou Njikam, Research Paper 133.
- Efficiency of Microenterprises in the Nigerian Economy, by Igbekele A. Ajibefun and Adebiyi G. Daramola, Research Paper 134.
- The Impact of Foreign Aid on Public Expenditure: The Case of Kenya, by James Njeru, Research Paper 135.
- The Effects of Trade Liberalization on Productive Efficiency: Electrical Industry in Cameroon, by Ousmanou Niikam. Research Paper 136.
- How Tied Aid Affects the Cost of Aid-Funded Projects in Ghana, by Barfour Osei, Research Paper 137.
- Exchange Rate Regimes and Inflation in Tanzania, by Longinus Rutasitara, Research Paper 138.
- Private Returns to Higher Education in Nigeria, by O.B.Okuwa, Research Paper 139.
- Uganda's Equilibrium Real Exchange Rate and Its Implications for Non-Traditional Export Performance, by Michael Atingi-Ego and Rachel Kaggwa Sebudde, Research Paper 140.

- Dynamic Inter-Links among the Exchange Rate, Price Level and Terms of Trade in a Managed Floating Exchange Rate System: The Case of Ghana, by Vijay K. Bhasin, Research Paper 141.
- Financial Deepening, Economic Growth and Development: Evidence from Selected Sub-Saharan African Countries, by John E. Udo Ndebbio, Research Paper 142.
- The Determinants of Inflation in South Africa: An Econometric Analysis, by Oludele A. Akinboade, Franz K. Siebrits and Elizabeth W. Niedermeier, Research Paper 143.
- The Cost of Aid Tying to Ghana, by Barfour Osei, Research Paper 144.
- A Positive and Normative Analysis of Bank Supervision in Nigeria, by A, Soyibo, S.O. Alashi and M.K. Ahmad, Research Paper 145.
- The Determinants of the Real Exchange Rate in Zambia, by Kombe O. Mungule, Research Paper 146.

  An Evaluation of the Viability of a Single Monetary Zone in ECOWAS, by Olawale Ogunkola, Research Paper 147.
- Analysis of the Cost of Infrastructure failures in a Developing Economy: The Case of Electricity Sector in Nigeria, by Adeola Adenikinju, Research Paper 148.

#### AFRICAN ECONOMIC RESEARCH CONSORTIUM

P.O. BOX 62882 - 00200 NAIROBI, KENYA

TELEPHONE (254-20) 2734150 2734153 2734157 2734163 2734166 2734179

FAX (254-20) 2734170 2734173

E-MAIL: communications @ aercafrica.org

WEB SITE: http://www.aercafrica.org

The principal objective of the African Economic Research Consortium (AERC), established in August 1988, is to strengthen local capacity for conducting independent, rigorous inquiry into problems pertinent to the management of economies in sub-Saharan Africa.

In response to special needs of the region, AERC has adopted a flexible approach to improve the technical skills of local researchers, allow for regional determination of research priorities, strengthen national institutions concerned with economic policy research, and facilitate closer ties between researchers and policy makers.

Since its establishment, AERC has been supported by private foundations, bilateral aid agencies and international organizations.

SPECIAL PAPERS contain the findings of commissioned studies in furtherance of AERC's programmes for research, training and capacity building.

RESEARCH PAPERS contain the edited and externally reviewed results of research financed by the AERC.

It is AERC's policy that authors of Special and Research papers are free to use material contained therein in other publications. Views expressed in such papers are those of the authors alone and should not be attributed to the AERC's sponsoring Members, Programme Committee, or Secretariat,

Further information concerning the AERC, and additional copies of Special and Research Papers, can be obtained by writing to: African Economic Research Consortium, P.O. Box 62882 - 00200, Nairobi, Kenya.

This work is licensed under a Creative Commons
Attribution - Noncommercial - NoDerivs 3.0 Licence.

To view a copy of the licence please see: http://creativecommons.org/licenses/by-nc-nd/3.0/